Author's response to reviews

Title: Henoch-Schonlein purpura with intracerebral haemorrhage in an adult patient: a case report

Authors:

Lazarus Karamadoukis (lazarus.karamadoukis@nbt.nhs.uk)
Linmarie Ludeman (linmarie.ludeman@glos.nhs.uk)
Anthony J Williams (tony.williams@glos.nhs.uk)

Version: 12 Date: 9 May 2008

Author's response to reviews:

Added the email of LL in title page.
In Case report the sentence "There was a significant deterioration of his serum creatinine from 108 to 152µmol/l." was changed to "There was a significant increase of his serum creatinine from 108 to 152µmol/l."